RESEARCH ARTICLE

## Longitudinal changes in tear cytokines and antimicrobial proteins in trachomatous disease

Amber Barton<sup>1\*</sup>, Nkoyo Faal<sup>2</sup>, Athumani Ramadhani<sup>1,3</sup>, Tamsyn Derrick<sup>1</sup>, Elias Mafuru<sup>3</sup>, Tara Mtuy<sup>1,3</sup>, Patrick Massae<sup>3</sup>, Aiweda Malissa<sup>3</sup>, Hassan Joof<sup>2</sup>, Pateh Makalo<sup>2</sup>, Ansumana Sillah<sup>4</sup>, Anna Harte<sup>1</sup>, Harry Pickering<sup>1</sup>, Robin Bailey<sup>1</sup>, David CW Mabey<sup>1</sup>, Matthew J. Burton<sup>1</sup>, Martin J. Holland<sup>1</sup>

1 Clinical Research Department, Faculty of Infectious and Tropical Diseases, London School of Hygiene and Tropical Medicine, Keppel Street, London, United Kingdom, 2 Medical Research Council Gambia at LSHTM, Atlantic Boulevard, Fajara, Banjul, The Gambia, 3 Department of Ophthalmology, Kilimanjaro Christian Medical Centre, Moshi, Tanzania, 4 National Eye Health Programme, Ministry of Health, Banjul, The Gambia

\* ab61@sanger.ac.uk

## Abstract

### Background

Trachoma is a neglected tropical disease caused by ocular infection with *Chlamydia trachomatis*, where repeated infections and chronic inflammation can ultimately result in scarring, trichiasis and blindness. While scarring is thought to be mediated by a dysregulated immune response, the kinetics of cytokines and antimicrobial proteins in the tear film have not yet been characterised.

## Methodology

Pooled tears from a Gambian cohort and Tanzanian cohort were semi-quantitatively screened using a Proteome Profiler Array to identify cytokines differentially regulated in disease. Based on this screen and previous literature, ten cytokines (CXCL1, IP-10, IFN- $\gamma$ , IL-1 $\beta$ , IL-8, IL-10, IL-12 p40, IL-1RA, IL-1 $\alpha$  and PDGF), lysozyme and lactoferrin were assayed in the Tanzanian cohort by multiplex cytokine assay and ELISA. Finally, CXCL1, IP-10, IL-8, lysozyme and lactoferrin were longitudinally profiled in the Gambian cohort by multiplex cytokine assay and ELISA.

### Results

In the Tanzanian cohort, IL-8 was significantly increased in those with clinically inapparent infection (p = 0.0086). Lysozyme, IL-10 and chemokines CXCL1 and IL-8 were increased in scarring (p = 0.016, 0.046, 0.016, and 0.037). CXCL1, IP-10, IL-8, lysozyme and lactoferrin were longitudinally profiled over the course of infection in a Gambian cohort study, with evidence of an inflammatory response both before, during and after detectable infection. CXCL1, IL-8 and IP-10 were higher in the second infection episode relative to the first (p = 0.0012, 0.044, and 0.04).



## G OPEN ACCESS

**Citation:** Barton A, Faal N, Ramadhani A, Derrick T, Mafuru E, Mtuy T, et al. (2023) Longitudinal changes in tear cytokines and antimicrobial proteins in trachomatous disease. PLoS Negl Trop Dis 17(10): e0011689. https://doi.org/10.1371/ journal.pntd.0011689

Editor: Stuart D. Blacksell, Mahidol Univ, Fac Trop Med, THAILAND

Received: February 20, 2023

Accepted: September 28, 2023

Published: October 20, 2023

**Copyright:** © 2023 Barton et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All data are in the in the manuscript and in <u>supporting information</u> files.

Funding: This work was funded by the RSTMH Small Grant (AB, rstmh.org). The two cohort studies were funded by the Wellcome Trust (098481/Z/12/Z to MJB, and 093368/Z/10/Z to DCWM, RB, MJB and MJH, wellcome.org), MRC UK (G982636/48103 to DCWM and RB, ukri.org/ councils/mrc) and EU Horizon 2020 grant programme (733373 to MJB and MJH, researchand-innovation.ec.europa.eu). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

#### Conclusions

These findings suggest that the ocular immune system responds prior to and continues to respond after detectable *C. trachomatis* infection, possibly due to a positive feedback loop inducing immune activation. Levels of CXC chemokines in successive infection episodes were increased, which may offer an explanation as to why repeated infections are a risk factor for scarring.

#### Author summary

The neglected tropical disease trachoma is caused by repeated ocular infections with *Chlamydia trachomatis*. Progression from infection to inflammation and scarring is thought to be due to a pathogenic immune response. This study aimed to identify new associations between tear proteins and trachoma, and to study how they change over time in those with infection. We found that inflammatory chemokines and the antibacterial enzyme lysozyme were increased in scarring, while IL-8 was increased during infection. In a separate cohort study, the same inflammatory chemokines were found to increase before and after detectable infection, and were higher in repeated infections.

#### Introduction

Ocular infection with *Chlamydia* (*C.*) *trachomatis* serovars A-C results in the neglected tropical disease trachoma, characterised by follicular conjunctivitis or papillary hypertrophy on the upper tarsal conjunctiva (active trachoma) [1]. However, many infections have no clinical signs [2]. Due to incomplete immunity and repeated exposure, recurrent episodes of infection and disease are common in endemic areas [3]. This chronic inflammation can result in scarring, and in severe cases progress to trichiasis and blindness. In 2021, 1.8 million people were estimated to be going blind as a result of trachoma [4] whilst 125 million lived in areas warranting trachoma control via antibiotics, facial cleanliness or environmental improvement [5].

The pathogenesis of trachoma is thought to be mediated by a dysregulated immune response [3], where infection-induced cytokines recruit tissue-damaging immune cells such as neutrophils and CD8<sup>+</sup> T cells. Gene expression of multiple cytokines is raised in the conjunctiva during each stage of disease: IL-1 $\beta$ , TGF- $\beta$ , TNF- $\alpha$ , IFN- $\gamma$ , IL-2, IL-4, IL-12 p40 during infection; IL-1 $\beta$ , TGF- $\beta$ , TNF- $\alpha$ , IL-17A, CCL18 and CXCL5 during active disease, and IL-1 $\beta$ , TGF- $\beta$  and TNF- $\alpha$  in scarring [6–9]. However as cytokine production is regulated at the post-transcriptional and translational level [10], conjunctival gene expression does not necessarily reflect the levels of secreted cytokines.

Tear fluid is increasingly emerging as a non-invasive source of biomarkers for both ocular and systemic disease [11]. Produced by the lacrimal glands, tears contain over a thousand proteins, including high levels of the antimicrobial proteins lysozyme and lactoferrin [12]. Tears also contain cytokines produced by local epithelial and immune cells, with two studies previously finding TNF- $\alpha$  to be raised in the tears of those with trachomatous scarring [13,14].

While a number of cytokines and proteins have been identified as raised or suppressed in trachomatous disease, in the absence of longitudinal data it is difficult to discern how these changes affect disease progression. Furthermore, given the vast number of proteins in tears, there is scope for as yet undiscovered relationships between tear composition and disease.

Studies to date have also focused on active disease and trachomatous scarring rather than inapparent *C. trachomatis* infection. We hypothesised that as yet unidentified cytokines may be associated with disease, and that longitudinal changes in tear cytokines during asymptomatic infections may relate to the downstream pathogenesis of trachoma.

We used proteome profiler arrays to semi-quantitatively screen for novel cytokines associated with different stages of trachoma. We then investigated these relationships in a Tanzanian cohort study, and tested whether tear lysozyme and lactoferrin are differentially regulated in trachoma. Finally, using tears from a second cohort study from The Gambia, we tracked changes in tear cytokines over the course of infection and disease.

#### Methods

#### Ethics statement

For the Gambian cohort, study procedures were approved by The Gambia Government/Medical Research Council, The Gambia Unit Joint Ethics Committee and the London School of Hygiene & Tropical Medicine Ethics Committee. For the Tanzanian cohort, study procedures were approved by the Tanzania National Institute for Medical Research, Kilimanjaro Christian Medical University College and London School of Hygiene & Tropical Medicine ethics committees. For both cohorts written informed consent was obtained from all study participants' guardians on their behalf.

#### Studies

Tear samples were collected from two longitudinal cohort studies that are detailed elsewhere [15,16]: briefly tears were included from fifteen time points in a Gambian cohort (bi-monthly over 6 months), and at two timepoints in a Tanzanian cohort (annually at months 72 and 84 post-baseline) (Table 1).

Cohort	Gambian		Tanzanian	
Age	4–15 years (entire cohort and in samples assayed)		11–19 years (entire cohort and in samples assayed)	
Participants	345		666	
	Every 2 weeks for 6 monthsEvery 3 months for 4 years then at 172 and 84 post baseline		years then at months ne	
Tear samples assayed	<ul> <li>454 total</li> <li>61 participants at on</li> <li>44 at two timepoints, 2</li> <li>24 at four, 6 at five, 7 a</li> <li>seven, 2 at eight, 1 at t</li> <li>155 samples from du</li> <li>infection, 28 active dis</li> <li>scarring</li> </ul>	e timepoint, 23 at three, at six, 6 at en rring sease and 31	<ul> <li>152 total</li> <li>88 participants at one timepoint, 31 participants at two timepoints</li> <li>15 samples from during infection, 2 active disease and 54 scarring</li> </ul>	
Grading	WHO simplified system		WHO FPC system	
	Active disease	TF	Active disease F2 or F3 P3	
		TI		Р3
	Scarring	TS	Scarring	C1, C2 or C3
		TT		
Active disease prevalence	24% (baseline timepoint)		5% (72 months post-recruitment)	
Scarring prevalence	5% (baseline timepoint)		26% (72 months post-recruitment)	

Table 1. Summary of the two cohorts from which tear samples were collected.

https://doi.org/10.1371/journal.pntd.0011689.t001

The Gambian cohort consisted of 345 children age 4–15 years recruited from nine villages. Participants were followed up every two weeks for six months [9,15]. At each visit children were clinically graded for trachoma using the WHO simplified system. Here the grades "Tra-chomatous inflammation—follicular (TF)" and "Trachomatous inflammation—intense (TI)" are considered "active disease", and the grades "Trachomatous scarring (TS)" and "Trachomatous trichiasis (TT)" as "scarring". Conjunctival swabs were collected from the right eye and stored in RNAlater (ThermoFisher) for subsequent *C. trachomatis* detection by polymerase chain reaction (PCR), as described previously [15]. Tears were collected at every timepoint by holding a sponge-tipped eye spear (Merocel, previously found to have highest recovery of tear cytokines relative to other ophthalmic sponges [17]) to the inferior conjunctival fornix for 30 seconds. The spear was then placed in a tube containing a protease inhibitor cocktail in 300µl PBS (Sigma) and stored at -70°C.

The Tanzanian cohort consisted of 666 children from three neighbouring communities, age 4–10 years at time of recruitment, and 11–19 years at the final timepoint. Participants were initially followed up every three months for four years, then at 72 and 84 months post-baseline recruitment. Tears were collected at the final two time points only. At each visit participants were clinically graded using the WHO FPC (Follicles, papillary hypertrophy, cicatriciae) system. Here a grade of F2, F3 or P3 is considered "active disease", and C1-3 considered "scarring". Conjunctival swabs were collected from the left eye and stored in RNAlater (ThermoFisher) for subsequent *C. trachomatis* detection by PCR [16]. Tears were collected using a standardised collection by Schirmer strips, allowing simultaneous measurement of dry eye (tear volume <15mm). Schirmer strips were stored at -70°C. After thawing, strips were placed in a 0.5ml microcentrifuge tube with a punctured base. The 0.5ml tube was placed in a 2ml microcentrifuge tube, and 20µl sterile PBS applied to the middle of the strip. Tears were finally eluted by centrifuging for 10 minutes at 15,700g in a micro-centrifuge chilled to 4°C.

#### Proteome profiler array

Two proteome profiler human cytokine array kits (R&D systems, ARY005B) were used to semi-quantitively screen for cytokines of interest. For the Gambian cohort four pools were created, each containing six 100µl tear samples made up to 1.5ml using the kit array buffer provided. The pools contained: (1) samples from healthy participants, (2) samples where *C. trachomatis* infection was detected, (3) samples from participants with conjunctival scarring, (4) samples from participants with both *C. trachomatis* infection and conjunctival scarring. For the Tanzanian cohort four pools were created with each containing six 10µl tear samples made up to 1.5ml with kit array buffer. The pools contained samples from: (1) healthy participants with no history of active disease; (2) healthy participants with a history of active disease (one of more episodes of active disease preceding but not at the time of sample collection); (3) participants with conjunctival scarring and no history of active disease, and (4) conjunctival scarring with trichiasis and a history of active disease.

The assay procedure was carried out according to manufacturer's instructions. Briefly,  $15\mu$ l detection antibody cocktail was added to each pool and incubated for one hour. Meanwhile, four nitrocellulose membranes containing 36 capture antibodies (against C5a, IL-4, IL-27, CD40L, IL-5, IL-32 $\alpha$ , G-CSF, IL-6, IP-10, GM-CSF, IL-8, CXCL11, CXCL1, IL-10, CCL2, CCL1, IL-12 p70, MIF, ICAM-1, IL-13, MIP-1, IFN- $\gamma$ , IL-16, RANTES, IL-1 $\alpha$ , IL-17, CXCL12, IL-1 $\beta$ , IL-17E, Serpin E1, IL-1RA, IL-18, TNF- $\alpha$ , IL-2, IL-21 and TREM-1) in duplicate were incubated in blocking buffer for one hour on a rocking platform shaker. Blocking buffer was aspirated from the membranes, and the detection antibody/pool mixture added. Membranes were incubated overnight at 4°C on a rocking platform shaker, then washed before incubation

with Streptavidin-Horseradish Peroxidase (HRP) for 30 minutes. Membranes were once again washed and Chemi Reagent mix added. Chemiluminesence was imaged using a ChemiDoc for the Gambian cohort, and a C600 Azzure for the Tanzanian cohort. ImageJ was used to find the integrated pixel density of each spot. The mean of each duplicate spot was taken, and normalised relative to the integrated pixel density of negative control spots. As arrays were semi-quantitative and performed using pooled samples without biological replicates, no formal hypothesis-testing was carried out.

#### Tear protein concentration

Total tear protein concentration was measured using a bicinchoninic acid assay (Pierce BCA Protein Assay Kit, ThermoFisher). Tears from the Tanzanian cohort were diluted 1:10 prior to protein quantitation, and protein quantitation carried out in singlicate. Protein concentration was measured in undiluted tears from the Gambian cohort in singlicate. 10µl tear samples were pipetted into a 96 well flat-bottomed plate. 200µl working reagent was added and the plate incubated for 30 minutes at 37°C. Absorbance was then read on a plate reader at 562nm. Concentrations were estimated from the standard curve using the drc package in R. For the Tanzanian cohort duplicates were averaged.

#### Multiplex cytokine assays

Cytokine concentrations were measured using a MILLIPLEX Human Cytokine/Chemokine Magnetic Bead Panel (Millipore). For the Tanzanian cohort, a 10-plex (CXCL1, IL-10, IL-1 $\alpha$ , IL-1 $\beta$ , IL-8, IP-10, IL-12 p40, IL1-RA, IFN- $\gamma$  & PDGF-AB/BB) assay was carried out in duplicate on tears diluted 1:10 in assay buffer. For the Gambian cohort, a 3-plex (CXCL1, IL-8 and IP-10) was carried out on undiluted tears in singlicate. Samples from the same participant at different timepoints were assayed on the same plate. Concentrations falling below the range of the standard curve were set to zero (19.2% of measurements in the Tanzanian cohort; 0.1% in the Gambian cohort). One sample in the Tanzanian cohort had insufficient beads and was excluded. Where the median fluorescence intensity was too high to estimate concentration, the concentration was set to that of the highest standard for that cytokine (0.5% of measurements in the Tanzanian cohort; 1.0% in the Gambian cohort). For the Tanzanian cohort the mean of the two replicates was taken. Finally, cytokine concentration was normalised relative to total protein concentration.

#### Antimicrobial protein auantification

Tear lysozyme concentrations were measured using a Human Lysozyme enzyme-linked immunosorbent assay (ELISA) Kit (abcam, ab108880), and lactoferrin concentrations using a Human Lactoferrin ELISA Kit (Immunology Consultants Laboratory, E-80LF). For the Tanzanian cohort, tears were initially diluted 1:1000,000 for lysozyme quantification, and 1:1000,000 for lactoferrin quantification. Tears were assayed in duplicate according to manufacturers' instructions (n = 152 samples for both lysozyme and lactoferrin). Any samples with an absorbance too high to estimate lysozyme concentration from the standard curve, or with an estimated lysozyme concentration > 10ng/ml diluted, were tested at a dilution of 1:10,000,000. For the Gambian cohort, tears were assayed in singlicate according to manufacturers' instructions (n = 454 samples for lactoferrin and n = 377 samples for lysozyme). Concentrations were estimated from the standard curve using the drc package in R. For the Tanzanian cohort duplicates were averaged.

#### Data analysis

Data analysis was carried out in the R statistical environment. For comparisons between different clinical groups where multiple samples from the same participant were included, a mixed effects logistic regression model (glmer function, lme4 package) was used to adjust for age, sex and multiple observations of the same participant. For comparisons between different clinical groups where each participant was only included once, a logistic regression model adjusting for age and sex was used. As data was not normally distributed, for paired tests a Wilcoxon signed rank test was used. P values below 0.05 were considered significant. Data was visualised using the ggplot2 package. In several figures data is visualised as a log-transformed fold change relative to the first infection time point. This timepoint was chosen as it was tested in more participants than any other timepoint, and allows the direction of change to be identified for each participant.

#### **Transcriptomic Re-analysis**

Transcriptomic datasets GSE20436, GSE20430 [18], GSE114556 [19], GSE105149 [20], GSE180238 [21] and GSE180027 [22] were downloaded from the NCBI Gene Expression Omnibus using the Bioconductor R package GEOquery. If not already, datasets were log-transformed, and quantile-normalised using the preprocessCore R package. Rows were collapsed to genes using the WGCNA package, taking the probe with the maximum mean expression as representative [23]. Differential expression analysis between conditions was carried out using the limma package. To compare gene expression between tissues, non-log transformed average expression was normalised relative to *ACTB*.

#### Results

# Tear inflammatory cytokines are raised in scarring in a Semi-quantitative screen of pooled tear samples

In the initial screen using pooled tear samples and semi-quantitative arrays, relative to the healthy control pools CXCL1, IP-10, IFN- $\gamma$ , IL-1 $\beta$ , IL-8, IL-10, IL-12 and IL-27 were higher in Gambian participants with scarring and in Tanzanian participants with scarring but no active disease history (Fig 1). Relative to healthy controls, IL-1 $\alpha$  was higher in Gambian participants with scarring, but not Tanzanian participants with scarring. IL-1RA was constitutively present



Fig 1. Mean integrated pixel density of each cytokine relative to negative control spots in each of eight pooled tear samples. Clinical status of each pool is indicated on the x axis. Cytokines are clustered by euclidean distance. As arrays were semi-quantitative and performed using pooled samples without biological replicates, no formal hypothesis-testing was carried out.

https://doi.org/10.1371/journal.pntd.0011689.g001

in tears at high levels. Based on this screen CXCL1, IP-10, IFN- $\gamma$ , IL-1 $\beta$ , IL-8, IL-10, IL-12 p40, IL1-RA and IL-1 $\alpha$  were selected for follow up multiplex cytokine assays on individual Tanzanian samples. IL-27 was not selected for further follow up due to lack of differential expression in microarray data [18]. PDGF-AB/BB, while not included in the semi-quantitative array, was selected for inclusion in the multiplex cytokine assay since it has previously been associated with conjunctival damage [24]. Lactoferrin and lysozyme concentrations were also assayed by ELISA as these had been suggested as markers of tear film health [25].

# Tear lysozyme and CXC chemokines are raised in scarring in a Tanzanian Cohort study

As tears were collected at two timepoints in the Tanzanian Cohort study (72 and 84 months post-recruitment), with some but not complete overlap in participants at each timepoint, a mixed effects logistic regression model was used to compare each group to healthy controls, adjusting for age, sex and multiple observations of the same participant. Compared with healthy controls (no infection, FPC clinical grades all zero), tear samples from Tanzanian cohort participants infected with *C. trachomatis* had higher levels of IL-8 (p = 0.0086, mixed effects logistic regression; Fig 2). In samples from those with scarring, lysozyme, IL-10 and CXC chemokines IL-8 and CXCL1 were significantly increased (p = 0.016, 0.046, 0.016, and 0.037 respectively, mixed effects logistic regression) while IP-10 and lactoferrin fell just short of significance (p = 0.093 and 0.074 respectively). While there were too few participants with active trachoma (clinical grades F > 1, P > 2, n = 2) to detect changes in active disease, IL-10,



Fig 2. Protein concentration for participants in different infection and clinical categories in the Tanzanian Cohort study. Comparisons with a p value < 0.1 for a mixed-effects logistic regression adjusting for age, sex and participant is shown. Comparisons with a p value < 0.05 are highlighted in bold. Each point represents a sample, coloured by cytokine/antimicrobial protein. Box plots indicate median and interquartile range (IQR), with whiskers extending to the most extreme values within 1.5 x IQR. Tear samples collected at both at 72 and 84 months post-recruitment were included.

https://doi.org/10.1371/journal.pntd.0011689.g002

IL-8 and CXCL1 were significantly increased in those with clinical grades F > 0 or P > 0 (p = 0.0031, 0.014, 0.037 respectively, logistic regression). Consistently, re-analysis of a gene expression array performed in a cross-sectional Gambian study independent of the samples assayed in this study [18] identified the genes encoding IL-10, IL-8 and CXCL1 as being upregulated (p < 0.05, linear modelling) during active disease (Table A in <u>S1 Text</u>).

# CXCL1 and lactoferrin are raised in Gambian participants with *C*. *trachomatis* infection

CXCL1, IP-10, IL-8, lysozyme and lactoferrin were selected for further profiling in a Gambian cohort study with more closely spaced and a greater number of timepoints available. Samples were selected to investigate the kinetics of infection and scarring (Table 2). Recovery of CXCL1, IP-10 and IL-8 from sponge-tipped eye spears in the Gambian cohort study was higher than from Schirmer strips in the Tanzanian study, while lysozyme and lactoferrin were lower (Fig A in S1 Text).

Analysing all timepoints together in an unpaired analysis, there was a marked trend towards higher CXCL1, IL-8, IP-10 and lactoferrin during infection. However, after adjusting for age, sex and repeated sampling of the same participant, only CXCL1 and lactoferrin reached significance (p = 0.032 and 0.019 respectively, mixed effects logistic regression; Fig 3). Active disease and scarring followed a similar pattern, with only an increase in lactoferrin during scarring reaching significance (p = 0.0074). For lysozyme on the other hand, there was a non-significant decrease in active disease (p = 0.096, logistic regression). Re-analysis of microarray data suggested that at the gene expression level, lysozyme is upregulated in the conjunctiva during active disease (p = 0.0001,  $log_2$ (fold change) = 0.5, linear modelling) while expression in the lacrimal gland is halved during inflammation (GSE105149, p = 0.1,  $log_2$ (fold change) = -1.0), suggesting the drop is likely mediated by reduced lacrimal gland secretion. In contrast to the Tanzanian Cohort study, lysozyme was unchanged in scarring (p = 0.2).

#### CXC inflammatory chemokines increase prior to detectable infection, peak four weeks after infection and are increased in successive infection episodes

Focusing only on the first episode of infection in the Gambian Cohort study, inflammatory CXC cytokines were raised four weeks after infection (p = 0.0095, 7.2 x  $10^{-5}$  and 0.0062 for

Subgroup	Timepoints assayed	Number of samples
Healthy throughout study (no positive PCRs or clinical signs throughout study)	One timepoint to act as control	43
Infection	First timepoint of infection	128
	Further timepoints during infection	27
	Timepoints leading up to infection	85
	Timepoints after infection	154
Scarring throughout the study period	Timepoint during scarring	2
Developed scarring during the study	Timepoints prior to scarring onset	35
	Timepoints after development of scarring	29
Active disease	Timepoints during active disease	28

Table 2. Tear samples selected from the Gambian cohort study to investigate the kinetics of infection and scarring.

https://doi.org/10.1371/journal.pntd.0011689.t002





CXCL1, IL-8 and IP-10 respectively, using a paired Wilcoxon signed-rank test to compare the normalised concentration four weeks post-infection relative to time of infection; Fig 4). Lyso-zyme however significantly dropped two weeks after infection (p = 0.013, paired Wilcoxon signed-rank test) while lactoferrin did not change. Compared with healthy controls who had no clinical signs or positive PCRs throughout the course of the study, CXCL1 and lactoferrin were raised two weeks before *C. trachomatis* infection was detected (p = 0.045 and 0.021, logistic regression model adjusting for age and sex; Fig 5).

A transcriptomic re-analysis was carried out to identify which tissues may be responsible for these changes (Table B in <u>S1 Text</u>). Lysozyme was highly expressed by the lacrimal gland, but also expressed by phagocytes and epithelial cells. CXC chemokines were expressed by epithelial and/or immune cells rather than the lacrimal gland (Table B in <u>S1 Text</u>), and were upregulated in response to in vitro *C. trachomatis* infection or LPS stimulation (Table C in <u>S1 Text</u>).

In participants with sustained infections lasting two or more weeks, chemokines and antimicrobial proteins remained constant during infection (Fig B in <u>S1 Text</u>). There were no significant changes in the timepoints leading up to active disease or scarring (Figs C and D in <u>S1</u> <u>Text</u>).



Fig 4. Change in normalised protein concentration relative to the infected time point for each participant in the Gambian Cohort study. Time points are separated by two weeks, and samples were limited to the first infection episode only. P values for a paired Wilcoxon signed-rank test comparing the normalised concentration at each time point relative to time of infection are shown, with p values < 0.05 highlighted in bold. Each point represents a sample, coloured by cytokine/antimicrobial protein. Grey lines show the log<sub>2</sub>(fold change) in concentration over time for each participant, while the black thick line shows the median log<sub>2</sub>(fold change).

In participants with successive infections, the second and third infection episodes were compared to the first infection episode (first time point of each episode only). CXCL1, IL-8 and IP-10 were raised in the second infection episode relative to the first (p = 0.0012, 0.044, and 0.04, paired Wilcoxon signed-rank test, n = 20; Fig 6).

#### Discussion

#### Kinetics of tear cytokine responses to C. trachomatis infection

These data represent the first report of cytokines and antimicrobial proteins longitudinally profiled in the tears of human participants over the course of ocular *C. trachomatis* infection and trachomatous disease. We found that CXC cytokines IL-8, IP-10 and CXCL1, and antimicrobial protein lactoferrin, were raised either during or following clinically inapparent infection. Consistently, following exposure to *C. trachomatis* the genes encoding IL-8 and IP-10 are upregulated by neutrophils [21], IL-8 and CXCL1 by macrophages [26], and CXCL1 by



**Fig 5.** Normalised protein concentrations prior to infection, compared with healthy controls. Protein concentrations normalised to total protein, in healthy control participants (individuals with no PCR positive infections or clinical signs throughout the entire study) compared with four weeks and two weeks prior to first detectable infection in individuals who did develop infection. Only the first episode of infection was analysed. P values for a logistic regression model comparing each timepoint with healthy controls, adjusting for age and sex, is shown. P values < 0.05 are highlighted in bold. Each point represents a sample, coloured by cytokine/antimicrobial protein. Box plots indicate median and IQR, with whiskers extending to the most extreme values within 1.5 x IQR.

https://doi.org/10.1371/journal.pntd.0011689.g005



**Fig 6.** Concentration of proteins normalised to total tear protein at the first timepoint of each infection episode. Healthy controls are shown for comparison. P values for a paired Wilcoxon signed-rank test comparing episodes 2 and 3 with episode 1 are shown, with p values < 0.05 highlighted in bold. Each point represents a sample, coloured by cytokine/antimicrobial protein. Box plots indicate median and IQR, with whiskers extending to the most extreme values within 1.5 x IQR.

epithelial cells [19]. IL-8, IP-10 and CXCL1 are all transcriptionally regulated by NF- $\kappa$ B [27], suggesting that cytokine expression is likely induced by pattern recognition receptors: either directly by innate recognition of *C. trachomatis* [28,29], or indirectly in response to *C. trachomatis*-mediated cell damage [30].

Whereas lactoferrin is anti-inflammatory and prevents *C. trachomatis* invasion in vitro [31], CXC cytokines are a double edged-sword, recruiting neutrophils and T cells to sites of infection [32] (Fig 7A). While CD4<sup>+</sup> T cell responses against *C. trachomatis* are necessary for resolution of infection, CD8<sup>+</sup> T cells and neutrophils may mediate the tissue damage seen in trachomatous scarring (Fig 7B) [3,33]. Animal models suggest that while CD8<sup>+</sup> T cells cause pathology through secretion of TNF- $\alpha$  [34,35], neutrophils may directly damage tissue through production of reactive oxygen species and matrix metalloproteinases [3,36]. In support of neutrophil recruitment playing a role in pathogenesis, a genetic variant resulting in lower levels of IL-8 increases resistance to trachomatous scarring [37].

Inflammatory cytokines can also inhibit lacrimal gland tear secretion [25,38]. This may explain why the antibacterial enzyme lysozyme, which is secreted by both the lacrimal gland and phagocytes in the conjunctiva, is suppressed two weeks after infection (p = 0.013) with a non-significant drop during active disease (p = 0.096) (Fig 7C). While lysozyme has not previously been measured in tears in relation to trachoma, a fall in levels has previously been observed in upper respiratory tract infections and dry eye disorders [25,39].

Interestingly, lactoferrin and CXCL1 were significantly higher than healthy controls two weeks prior to detectable infection (p = 0.045 and 0.021), suggesting an immune response is triggered well before *C. trachomatis* reaches a high enough bacterial load to be detected by 16s rRNA PCR. Furthermore, CXCL1, IP-10 and IL-8 peaked at around four weeks post-infection, even though by this time *C. trachomatis* was no longer detectable by PCR. One possibility is that after an infection has been resolved or suppressed, a positive feedback loop continues to propagate an inflammatory immune response [40]. Another possibility is that live *C. trachomatis* suppresses inflammatory responses during infection, and this suppression wanes once the bacteria are killed. For example, in vitro IP-10 levels inversely correlate with bacterial burden, suggesting active suppression by *C. trachomatis* [41]. It is also possible that while not detectable in the conjunctiva, *C. trachomatis* is present at a sub-detectable level, or persisting in other ocular tissues or the lacrimal gland. Finally, it may be that other bacteria contribute to inflammation following resolution of infection: for example, it has previously been found that



**Fig 7. A model for trachoma pathogenesis based on study findings.** a. Infection results in activation of pattern recognition receptors and therefore an increase in CXC chemokines, resulting in recruitment of T cells, Neutrophils and Macrophages b. CD8<sup>+</sup> T cells and neutrophils can mediate tissue damage, resulting in trachomatous scarring c. Inflammatory cytokines suppress lacrimal gland secretion, decreasing lysozyme tear concentration d. Immune cells recruited to the conjunctiva and/or immune memory result in enhanced inflammation in successive infection episodes e. At later stages of scarring lysozyme is raised, possibly due to secretion by recruited macrophages.

*Streptococcus pneumoniae* and *Haemophilus influenzae* are more commonly cultured from those with active trachoma than healthy controls [42,43].

We also found that CXC chemokine responses were higher in the second infection episode relative to the first episode. This could explain why repeated infection episodes increase the likelihood of scarring and may be due to several mechanisms. Firstly, it may be that the immune cells recruited to the conjunctiva during the first episode are still present, and therefore mount a stronger inflammatory response in subsequent episodes (Fig 7D). Secondly, activation of memory T cells could result in an enhanced secondary immune response. Finally, epigenetic changes in immune cells could result in a "trained" innate immune response during subsequent infection episodes [44].

Our results contrasted to others in that unlike previous gene expression analyses, which found that infection induced IFN- $\gamma$  in the conjunctiva [6,8,9,15], our assay was not sufficiently sensitive to consistently detect IFN- $\gamma$  in the tears. Furthermore, we did not detect a significant change in IL-12 or IL-1 $\beta$  during infection [6,8,9].

#### Cytokines in active disease and scarring

No significant (p < 0.05) changes in cytokines were detected during active disease, likely due to low numbers of samples in the Tanzanian cohort, and selection of samples to focus on infection in the Gambian cohort. However, in the Tanzanian cohort IL-10, IL-8 and CXCL1 were raised in those with clinical grades F > 0 or P > 0 (p < 0.05), while in the Gambian cohort there was a non-significant increase in IL-8 and CXCL1 in active disease (p < 0.15). This is consistent with transcriptomic arrays finding an increase in conjunctival IL-10, IL-8 and CXCL1 mRNA [18], quantitative PCR finding an increase in conjunctival IL-10 mRNA [8],

and cytokine assays finding an increase in tear IL-8 and IL-10 [14] during active disease. A variety of cells could be responsible for secreting these cytokines: for example, monocytes, dendritic cells and B cells have each been identified as potential sources of IL-10 following *Chlamydia* infection [45–47]. Furthermore, conjunctival *FOXP3* expression is increased in active trachoma, suggesting that regulatory T cells might be present and contribute to tear IL-10 [15].

In the Tanzanian cohort CXCL1, IL-10 and IL-8 were significantly increased in scarring (p < 0.05). CXCL1 and IL-8 were also higher in Gambian cohort samples with scarring, but this difference was not significant after adjusting for repeated measures. These findings are in contrast to Skwor et al. 2008 [14], which found that IL12 and IL-1 $\beta$  were elevated in scarring, while IL-10 and IL-8 were unchanged. However, both suggest the ocular mucosa remains in an inflammatory state during scarring. In the Tanzanian study lysozyme was significantly increased (p = 0.016) in scarring, whereas in the Gambian cohort study lysozyme was unchanged. This difference could be due to the children in the Tanzanian study being older, and therefore at a later scarring stage where phagocyte secretion contributes more to tear lysozyme than the lacrimal gland (Fig 7E). Lactoferrin was significantly raised in scarring in the Gambian cohort (p = 0.0074) and was raised but fell short of significance in the Tanzanian cohort (p = 0.074). While this is likely due to sustained inflammation, it is not yet known whether lactoferrin plays any causative role in progressing or attenuating scarring trachoma.

#### Limitations

This study was limited in that the two cohort studies assayed here are not directly comparable. Firstly, the Gambian cohort were younger, and therefore had a higher prevalence of infection and lower prevalence of scarring. Furthermore, while in the Tanzanian cohort infection was detected by a multiplex DNA PCR targeting genes omcB and pORF2, in the Gambian cohort infection was detected by a Chlamydial 16s rRNA PCR. Although both methods have >90% specificity when compared to commercial kits, the multiplex DNA PCR has 90% sensitivity and the 16s RNA PCR 75% sensitivity [48,49]. Tears were also collected by two different methods: surgical eye sponges in the Gambian study, and Schirmer strips in the Tanzanian study, resulting in differences in cytokine and antimicrobial protein yield (Fig A in S1 Text). Further impact from method of collection, time in storage and additives cannot be completely excluded. However, to avoid protein degradation, all samples were collected on ice in the field and stored at -70°C; Gambian samples were stored with the addition of a cocktail of protease inhibitors for long term storage; and samples selected had not been subjected to freeze-thaw prior to use on the array profiler and multiplex cytokine assays.

#### Conclusions

Through a semi-quantitative screen we identified ten cytokines of interest for further follow up. IL-8 was raised in *C. trachomatis* infection, and multiple cytokines and lysozyme raised in conjunctival scarring. CXCL1, IP-10, IL-8, lysozyme and lactoferrin were longitudinally profiled over the course of infection in a separate cohort, with evidence of an inflammatory response both before and after detectable infection. Increased inflammation was observed in later infection episodes. This could have implications in vaccine development, suggesting that natural immune responses contribute to rather than protect against pathogenesis.

### **Supporting information**

**S1 Text.** Fig A. Comparison of protein recovery from Schirmer strips in the Tanzanian cohort study with recovery from sponge-tipped eye spears in the Gambian cohort study. Both analyte concentration (ng/ml) and analyte concentration normalised to total protein (ng/ mg total protein) are shown. Points each represent one sample, coloured by whether they represent a healthy control or other sample. Box plots indicate median and IQR, with whiskers extending to the most extreme values within 1.5 x IQR.Fig B. Protein concentration at the first infected time point, two weeks after infection and four weeks after infection in participants with sustained infections. Paired tests were carried out on time points from the same infection episode only. P values for a paired Wilcoxon signed-rank test comparing the normalised concentration at each time point relative to the first time point of infection are shown. Each point represents a sample. Grey lines show the change in concentration over time for each participant and infection episode, while the black thick line shows the change in the median. Fig C. Change in normalised protein concentration relative to the first scarring time point for each participant. Samples were limited to the first scarring episode only. P values for a paired Wilcoxon signed-rank test comparing the normalised concentration at each time point relative to time of incident scarring are shown. Each point represents a sample, coloured by cytokine/antimicrobial protein. Grey lines show the log<sub>2</sub>(fold change) in concentration over time for each participant, while the black thick line shows the median log<sub>2</sub>(fold change). Fig D. Change in normalised protein concentration relative to the first active disease time point for each participant. Samples were limited to those without any scarring or infection. P values for a paired Wilcoxon signed-rank test comparing the normalised concentration at each time point relative to time of active disease onset are shown. Each point represents a sample, coloured by cytokine/antimicrobial protein. Grey lines show the  $\log_2(\text{fold change})$  in concentration over time for each participant, while the black thick line shows the median log<sub>2</sub>(fold change). Table A. Differential expression of IL10, IL1B, IL8 and CXCL1 in the conjunctiva in active trachomatous disease relative to healthy controls in datasets GSE20436 and GSE20430. Samples were taken from a Gambian cross-sectional case-control study. Gene expression was quantile normalised and probes were collapsed to genes, taking the probe with the maximum mean as being representative. Differential expression analysis was carried out using the Bioconductor R package limma. Table B.Mean expression of CXCL1, IL8, CXCL10 (IP10), LYZ and LTF relative to ACTB, in epithelial cells, neutrophils, monocytes, macrophages, activated macrophages and the lacrimal gland. Datasets GSE114556, GSE105149, GSE180238 and GSE180027 were downloaded, quantile normalised and probes were collapsed to genes, taking the probe with the maximum mean as being representative. The mean expression of each gene (non-log transformed) was normalised relative to ACTB expression. Table C. Differential expression of CXCL1, IL8, CXCL10 (IP10), LYZ and LTF following in vitro exposure of epithelial cells, neutrophils or macrophages to either C. trachomatis or LPS. Datasets GSE114556, GSE180238 and GSE180027 were downloaded, quantile normalised and probes were collapsed to genes, taking the probe with the maximum mean as being representative. Differential expression analysis was carried out using the Bioconductor R package limma. (DOCX)

**S1 Dataset. Data generated by this study.** Integrated pixel densities normalised to negative control spots for the proteome profiler array screen; target analyte and total protein concentrations for each sample in the Tanzanian cohort study; and target analyte and total protein concentrations for each sample in the Gambian cohort study. (XLSX)

#### Acknowledgments

We would like to thank the study participants, as well as the trachoma field and technical teams.

#### **Author Contributions**

Conceptualization: Amber Barton, Martin J. Holland.

Formal analysis: Amber Barton.

- **Funding acquisition:** Amber Barton, Robin Bailey, David CW Mabey, Matthew J. Burton, Martin J. Holland.
- **Investigation:** Amber Barton, Nkoyo Faal, Athumani Ramadhani, Tamsyn Derrick, Elias Mafuru, Tara Mtuy, Patrick Massae, Aiweda Malissa, Hassan Joof, Pateh Makalo, Ansumana Sillah, Anna Harte, Harry Pickering, Robin Bailey, David CW Mabey, Matthew J. Burton, Martin J. Holland.

Supervision: Martin J. Holland.

Visualization: Amber Barton.

Writing - original draft: Amber Barton.

Writing – review & editing: Nkoyo Faal, Athumani Ramadhani, Tamsyn Derrick, Elias Mafuru, Tara Mtuy, Patrick Massae, Aiweda Malissa, Hassan Joof, Pateh Makalo, Ansumana Sillah, Anna Harte, Harry Pickering, Robin Bailey, David CW Mabey, Matthew J. Burton, Martin J. Holland.

#### References

- Ramadhani AM, Derrick T, Holland MJ, Burton MJ. Blinding Trachoma: Systematic Review of Rates and Risk Factors for Progressive Disease. PLoS Negl Trop Dis. 2016; 10(8):1–19. <u>https://doi.org/10. 1371/journal.pntd.0004859</u> PMID: 27483002
- Hu VH, Harding-Esch EM, Burton MJ, Bailey RL, Kadimpeul J, Mabey DCW. Epidemiology and control of trachoma: Systematic review. Trop Med Int Heal. 2010; 15(6):673–91. https://doi.org/10.1111/j.1365-3156.2010.02521.x PMID: 20374566
- Hu VH, Holland MJ, Burton MJ. Trachoma: Protective and Pathogenic Ocular Immune Responses to Chlamydia trachomatis. PLoS Negl Trop Dis. 2013; 7(2). https://doi.org/10.1371/journal.pntd.0002020 PMID: 23457650
- Solomon Anthony W., Burton MJ, Gower EW, Harding-Esch EM Oldenburg CE, Taylor HR, et al. Trachoma. Nat Rev Dis Prim. 2022; 8(32).
- WHO. WHO Alliance for the Global Elimination of Trachoma: progress report on elimination of trachoma, 2021. Vol. 98, Weekly epidemiological record. 2022.
- Bobo L, Novak N, Mkocha H, Vitale S, West S, Quinn TC. Evidence for a predominant proinflammatory conjunctival cytokine response in individuals with trachoma. Infect Immun. 1996; 64(8):3273–9. <a href="https://doi.org/10.1128/iai.64.8.3273-3279.1996">https://doi.org/10.1128/iai.64.8.3273-3279.1996</a> PMID: 8757864
- Burton MJ, Ramadhani A, Weiss HA, Hu V, Massae P, Burr SE, et al. Active trachoma is associated with increased conjunctival expression of IL17A and profibrotic cytokines. Infect Immun. 2011; 79 (12):4977–83. https://doi.org/10.1128/IAI.05718-11 PMID: 21911461
- Burton MJ, Bailey RL, Jeffries D, Mabey DCW, Holland MJ. Cytokine and fibrogenic gene expression in the conjunctivas of subjects from a Gambian community where trachoma is endemic. Infect Immun. 2004; 72(12):7352–6. https://doi.org/10.1128/IAI.72.12.7352-7356.2004 PMID: 15557667
- Faal N, Bailey RL, Sarr I, Joof H, Mabey DCW, Holland MJ. Temporal cytokine gene expression patterns in subjects with trachoma identify distinct conjunctival responses associated with infection. Clin Exp Immunol. 2005; 142(2):347–53. https://doi.org/10.1111/j.1365-2249.2005.02917.x PMID: 16232223
- Anderson P. Post-transcriptional control of cytokine production. Nat Immunol. 2008; 9(4):353–9. <a href="https://doi.org/10.1038/ni1584">https://doi.org/10.1038/ni1584</a> PMID: 18349815
- Pieragostino D, D'Alessandro M, di Ioia M, Di Ilio C, Sacchetta P, Del Boccio P. Unraveling the molecular repertoire of tears as a source of biomarkers: Beyond ocular diseases. Proteomics—Clin Appl. 2015; 9(1–2):169–86. https://doi.org/10.1002/prca.201400084 PMID: 25488355
- Zhou L, Beuerman RW. Tear analysis in ocular surface diseases. Prog Retin Eye Res. 2012; 31 (6):527–50. https://doi.org/10.1016/j.preteyeres.2012.06.002 PMID: 22732126

- Conway DJ, Holland MJ, Bailey RL, Campbell AE, Mahdi OSM, Jennings R, et al. Scarring trachoma is associated with polymorphism in the tumor necrosis factor alpha (TNF-α) gene promoter and with elevated TNF-α levels in tear fluid. Infect Immun. 1997; 65(3):1003–6.
- Skwor TA, Atik B, Kandel RP, Adhikarl HK, Sharma B, Dean D. Role of secreted conjunctival mucosal cytokine and chemokine proteins in different stages of trachomatous disease. PLoS Negl Trop Dis. 2008;2(7). https://doi.org/10.1371/journal.pntd.0000264 PMID: 18628987
- Faal N, Bailey RL, Jeffries D, Joof H, Sarr I, Laye M, et al. Conjunctival FOXP3 expression in trachoma: Do regulatory T cells have a role in human ocular Chlamydia trachomatis infection? PLoS Med. 2006; 3 (8):1292–301. https://doi.org/10.1371/journal.pmed.0030266 PMID: 16881731
- Ramadhani AM, Derrick T, MacLeod D, Massae P, Mafuru E, Malisa A, et al. Progression of scarring trachoma in Tanzanian children: A four-year cohort study. PLoS Neglected Tropical Diseases. 2019; 13 (8):1–16. https://doi.org/10.1371/journal.pntd.0007638 PMID: 31412025
- Inic-Kanada A, Nussbaumer A, Montanaro J, Belij S, Schlacher S, Stein E, et al. Comparison of ophthalmic sponges and extraction buffers for quantifying cytokine profiles in tears using Luminex technology. Mol Vis. 2012; 18(November):2717–25. PMID: 23233782
- Natividad A, Freeman TC, Jeffries D, Burton MJ, Mabey DCW, Bailey RL, et al. Human conjunctival transcriptome analysis reveals the prominence of innate defense in Chlamydia trachomatis infection. Infect Immun. 2010; 78(11):4895–911. https://doi.org/10.1128/IAI.00844-10 PMID: 20823212
- Ohmer M, Tzivelekidis T, Niedenführ N, Volceanov-Hahn L, Barth S, Vier J, et al. Infection of HeLa cells with Chlamydia trachomatis inhibits protein synthesis and causes multiple changes to host cell pathways. Cell Microbiol. 2019; 21(4):1–13. https://doi.org/10.1111/cmi.12993 PMID: 30551267
- Rosenbaum JT, Choi D, Harrington CA, Wilson DJ, Grossniklaus HE, Sibley CH, et al. Gene expression profiling and heterogeneity of nonspecific orbital inflammation affecting the lacrimal gland. JAMA Ophthalmol. 2017; 135(11):1156–62. <u>https://doi.org/10.1001/jamaophthalmol.2017.3458</u> PMID: 28975236
- Virok DP, Tömösi F, Keller-Pintér A, Szabó K, Bogdanov A, Poliska S, et al. Indoleamine 2,3-Dioxygenase Cannot Inhibit Chlamydia trachomatis Growth in HL-60 Human Neutrophil Granulocytes. Front Immunol. 2021; 12(November):1–14. https://doi.org/10.3389/fimmu.2021.717311 PMID: 34819931
- Murthy S, Karkossa I, Schmidt C, Hoffmann A, Hagemann T, Rothe K, et al. Danger signal extracellular calcium initiates differentiation of monocytes into SPP1/osteopontin-producing macrophages. Cell Death Dis. 2022; 13(1). https://doi.org/10.1038/s41419-022-04507-3 PMID: 35022393
- Langfelder P, Horvath S. WGCNA: An R package for weighted correlation network analysis. BMC Bioinformatics. 2008; 9.
- El-Asrar AMA, Geboes K, Tabbara KF, Al-Kharashi SA, Missotten L, Desmet V. Immunopathogenesis of conjunctival scarring in trachoma. Eye. 1998; 12(3):453–60. https://doi.org/10.1038/eye.1998.104 PMID: 9775249
- Hanstock HG, Edwards JP, Walsh NP. Tear lactoferrin and lysozyme as clinically relevant biomarkers of mucosal immune competence. Front Immunol. 2019; 10(MAY):1–11. <u>https://doi.org/10.3389/fimmu.</u> 2019.01178 PMID: 31231369
- Yeung ATY, Hale C, Lee AH, Gill EE, Bushell W, Parry-Smith D, et al. Exploiting induced pluripotent stem cell-derived macrophages to unravel host factors influencing Chlamydia trachomatis pathogenesis. Nat Commun. 2017; 8:1–12.
- 27. Liu T, Zhang L, Joo D, Sun SC. NF-κB signaling in inflammation. Signal Transduct Target Ther. 2017; 2 (March).
- Nagarajan UM, Ojcius DM, Stahl L, Rank RG, Darville T. Chlamydia trachomatis Induces Expression of IFN-γ-Inducible Protein 10 and IFN-β Independent of TLR2 and TLR4, but Largely Dependent on MyD88. J Immunol. 2005; 175(1):450–60.
- Agrawal T, Bhengraj AR, Vats V, Salhan S, Mittal A. Expression of TLR 2, TLR 4 and iNOS in Cervical Monocytes of Chlamydia trachomatis-infected Women and Their Role in Host Immune Response. Am J Reprod Immunol. 2011; 66(6):534–43. <u>https://doi.org/10.1111/j.1600-0897.2011.01064.x</u> PMID: 21883620
- Yang C, Lei L, Collins JWM, Briones M, Ma L, Sturdevant GL, et al. Chlamydia evasion of neutrophil host defense results in NLRP3 dependent myeloid-mediated sterile inflammation through the purinergic P2X7 receptor. Nat Commun. 2021; 12(1):1–16.
- Sessa R, Di Pietro M, Filardo S, Bressan A, Rosa L, Cutone A, et al. Effect of bovine lactoferrin on Chlamydia trachomatis infection and inflammation. Biochem Cell Biol. 2017; 95(1):34–40. <u>https://doi.org/10.1139/bcb-2016-0049</u> PMID: 28094551

- Griffith JW, Sokol CL, Luster AD. Chemokines and chemokine receptors: Positioning cells for host defense and immunity. Annu Rev Immunol. 2014; 32:659–702. <u>https://doi.org/10.1146/annurevimmunol-032713-120145 PMID: 24655300</u>
- Lijek RS, Helble JD, Olive AJ, Seiger KW, Starnbach MN. Pathology after chlamydia trachomatis infection is driven by nonprotective immune cells that are distinct from protective populations. Proc Natl Acad Sci. 2018; 115(9):2216–21. https://doi.org/10.1073/pnas.1711356115 PMID: 29440378
- Murthy AK, Li W, Chaganty BKR, Kamalakaran S, Guentzel MN, Seshu J, et al. Tumor necrosis factor alpha production from CD8+ T cells mediates oviduct pathological sequelae following primary genital Chlamydia muridarum Infection. Infect Immun. 2011; 79(7):2928–35. <u>https://doi.org/10.1128/IAI.05022-11 PMID: 21536799</u>
- 35. Manam S, Thomas JD, Li W, Maladore A, Schripsema JH, Ramsey KH, et al. Tumor necrosis factor (TNF) receptor superfamily member 1b on CD8+ T cells and TNF receptor superfamily member 1a on non-CD8+ T cells contribute significantly to upper genital tract pathology following chlamydial infection. J Infect Dis. 2015; 211(12):2014–22. https://doi.org/10.1093/infdis/jiu839 PMID: 25552370
- Lee HY, Schripsema JH, Sigar IM, Murray CM, Lacy SR, Ramsey KH. A link between neutrophils and chronic disease manifestations of Chlamydia muridarum urogenital infection of mice. FEMS Immunol Med Microbiol. 2010; 59(1):108–16. https://doi.org/10.1111/j.1574-695X.2010.00668.x PMID: 20370824
- Natividad A, Hull J, Luoni G, Holland M, Rockett K, Joof H, et al. Innate immunity in ocular Chlamydia trachomatis infection: Contribution of IL8 and CSF2 gene variants to risk of trachomatous scarring in Gambians. BMC Med Genet. 2009; 10:1–11.
- Zoukhri D. Effect of inflammation on lacrimal gland function. Exp Eye Res. 2006; 82(5):885–98. <a href="https://doi.org/10.1016/j.exer.2005.10.018">https://doi.org/10.1016/j.exer.2005.10.018</a> PMID: 16309672
- Deluise VP, Tabbara KF. Quantitation of Tear Lysozyme Levels in Dry-Eye Disorders. Arch Ophthalmol. 1983; 101(4):634–5. https://doi.org/10.1001/archopht.1983.01040010634022 PMID: 6838425
- 40. Beutler B. Microbe sensing, positive feedback loops, and the pathogenesis of inflammatory diseases. Immunol Rev. 2009; 227(1):1–7. https://doi.org/10.1111/j.1600-065X.2008.00733.x PMID: 19120489
- Schott BH, Antonia AL, Wang L, Pittman KJ, Sixt BS, Barnes AB, et al. Modeling of variables in cellular infection reveals CXCL10 levels are regulated by human genetic variation and the Chlamydia-encoded CPAF protease. Sci Rep. 2020; 10(1):1–12.
- **42.** Burton MJ, Hu VH, Massae P, Burr SE, Chevallier C, Afwamba IA, et al. What Is causing active trachoma? The role of nonchlamydial bacterial pathogens in a low prevalence setting. Investig Ophthalmol Vis Sci. 2011; 52(8):6012–7.
- 43. Burr SE, Hart JD, Edwards T, Baldeh I, Bojang E, Harding-Esch EM, et al. Association between Ocular Bacterial Carriage and Follicular Trachoma Following Mass Azithromycin Distribution in The Gambia. PLoS Negl Trop Dis. 2013; 7(7). https://doi.org/10.1371/journal.pntd.0002347 PMID: 23936573
- 44. Netea MG, Domínguez-Andrés J, Barreiro LB, Chavakis T, Divangahi M, Fuchs E, et al. Defining trained immunity and its role in health and disease. Nat Rev Immunol. 2020; 20(6):375–88. <u>https://doi.org/10.1038/s41577-020-0285-6 PMID: 32132681</u>
- 45. Öhman H, Tiitinen A, Halttunen M, Birkelund S, Christiansen G, Koskela P, et al. IL-10 polymorphism and cell-mediated immune response to Chlamydia trachomatis. Genes Immun. 2006; 7(3):243–9. https://doi.org/10.1038/sj.gene.6364293 PMID: 16525502
- 46. Marks E, Tam MA, Lycke NY. The female lower genital tract is a privileged compartment with IL-10 producing dendritic cells and poor Th1 immunity following Chlamydia trachomatis infection. PLoS Pathog. 2010; 6(11). https://doi.org/10.1371/journal.ppat.1001179 PMID: 21079691
- 47. Sanchez LR, Godoy GJ, Serrán MG, Breser ML, Vernengo FF, Engel P, et al. IL-10 producing B cells dampen protective T cell response and allow chlamydia muridarum infection of the male genital tract. Front Immunol. 2019; 10(MAR):1–15. https://doi.org/10.3389/fimmu.2019.00356 PMID: 30881362
- Butcher R, Houghton J, Derrick T, Ramadhani A, Herrera B, Last AR, et al. Reduced-cost Chlamydia trachomatis-specific multiplex real-time PCR diagnostic assay evaluated for ocular swabs and use by trachoma research programmes. J Microbiol Methods. 2017; 139(January):95–102. https://doi.org/10. 1016/j.mimet.2017.04.010 PMID: 28487054
- 49. Faal NI. Conjunctival immune responses in human ocular chlamydial infections. 2011.